Author's response to reviews

Title: A demonstration project for the implementation of the WHO antenatal care model in Mozambique: A cluster randomized controlled trial: Study protocol

Authors:

Leonardo Chavane (leochavane@gmail.com)
Mario Merialdi (merialdim@who.int)
Ana Pilar Betran (betrana@who.int)
Jennifer Requejo-Harris (jrequejo@jhsp.h.edu)
Eduardo Bergel (bergel@gmail.com)
Alicia Aleman (aaleman@unicem-web.org)
Mercedes Colomar (mcolomar@unicem-web.org)
Maria Luisa Cafferata (mcafferata@unicem-web.org)
Alicia Carbonell (carbonella@who.int)
Beatrice Crahay (icrh.coord@tdm.co.mz)
Therese Delvaux (TDelvaux@itg.be)
Diederike Geelhoed (geelhoed.d.w@gmail.com)
Metin Gülmezoglu (gulmezoglug@gmail.com)
Celsa Regina Malapende (CelsaMalapende@cmam.gov.mz)
Armando Melo (lagicy@yahoo.com.br)
My Huong Nguyen (nguyenh@who.int)
Nafissa Bique Osman (osman@teledata.mz)
Mariana Widmer (widmerm@who.int)
Marleen Temmerman (temmermanm@who.int)
Fernando Althabe (falthabe@iecs.org.ar)

Version: 2
Date: 12 November 2013

Author's response to reviews:

Not applicable. Transfer from BMC Pregnancy and Childbirth